Author’s response to reviews

Title: Survival After Lobectomy Versus Sub-Lobar Resection in Elderly with Stage I NSCLC: A Meta-analysis

Authors:

Siyuan Dong (dsycmu@163.com)
Steven Andrew Roberts (sroberts22@mgh.harvard.edu)
Shuang Chen (loscs@126.com)
Xinwen Zhong (zhongxinwencmu@163.com)
Shize Yang (yangshizecmu@163.com)
Xiaohan Qu (quxiaohancmu@163.com)
Shun Xu (shunxucmu@163.com)

Version: 5 Date: 27 Mar 2019

Author’s response to reviews:

Dear Megan Barling:

Thank you for your letter concerning our manuscript. Those comments are all valuable and very helpful for revising and improving our paper, as well as the important guiding significance to our researches. We have studied comments carefully and tried our best to improve the manuscript which we hope meet with approval.

1. Copy Edit
   Thanks for your suggestion. We feel sorry for our poor writing. Our manuscript was revised by a native speaker to correct the language problem.

2. Overlap
   Thanks for your reminding. We have made extensive corrections to our previous draft.

3. Clean Manuscript
   We upload a single, final, clean version.

Once again, we appreciate for editor’s warm work earnestly. Thank you for giving me the opportunity
to revise and resubmit this manuscript. I look forward to work with you and the reviewer to move the manuscript closer to publication in the BMC Surgery.